

GOPEN ACCESS

Citation: Lydeamore MJ, Campbell PT, Price DJ, Wu Y, Marcato AJ, Cuningham W, et al. (2020) Estimation of the force of infection and infectious period of skin sores in remote Australian communities using interval-censored data. PLoS Comput Biol 16(10): e1007838. https://doi.org/ 10.1371/journal.pcbi.1007838

Editor: Roger Dimitri Kouyos, University of Zurich, SWITZERLAND

Received: July 22, 2019

Accepted: April 1, 2020

Published: October 5, 2020

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pcbi.1007838

Copyright: © 2020 Lydeamore et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The software developed to perform analysis in this manuscript is available at https://github.com/MikeLydeamore/tmi.

RESEARCH ARTICLE

Estimation of the force of infection and infectious period of skin sores in remote Australian communities using intervalcensored data

Michael J. Lydeamore ^{1,2*}, Patricia T. Campbell ^{3,4}, David J. Price ^{3,5}, Yue Wu⁶, Adrian J. Marcato ³, Will Cuningham ⁷, Jonathan R. Carapetis ^{6,8}, Ross M. Andrews^{7,9}, Malcolm I. McDonald ¹⁰, Jodie McVernon ^{3,4,5}, Steven Y. C. Tong ^{3,7}, James M. McCaw ^{1,3,5}

 School of Mathematics and Statistics, The University of Melbourne, Melbourne, Australia, 2 Department of Infectious Diseases, The Alfred and Central Clinical School, Monash University, Melbourne, Australia, 3 Peter Doherty Institute for Infection and Immunity, The Royal Melbourne Hospital and The University of Melbourne, Melbourne, Australia, 4 Murdoch Children's Research Institute, The Royal Children's Hospital, Melbourne, Australia, 5 Melbourne School of Population and Global Health, The University of Melbourne, Melbourne, Australia, 6 Telethon Kids Institute, University of Western Australia, Perth, Australia, 7 Menzies School of Health Research, Charles Darwin University, Darwin, Australia, 8 Perth Children's Hospital, Perth, Australia, 9 National Centre for Epidemiology & Population Health, Australian National University, Canberra, Australia, 10 Australian Institute of Tropical Health and Medicine, James Cook University, Cairns, Australia

* michael.lydeamore@monash.edu

Abstract

Prevalence of impetigo (skin sores) remains high in remote Australian Aboriginal communities, Fiji, and other areas of socio-economic disadvantage. Skin sore infections, driven primarily in these settings by Group A Streptococcus (GAS) contribute substantially to the disease burden in these areas. Despite this, estimates for the force of infection, infectious period and basic reproductive ratio-all necessary for the construction of dynamic transmission models—have not been obtained. By utilising three datasets each containing longitudinal infection information on individuals, we estimate each of these epidemiologically important parameters. With an eye to future study design, we also quantify the optimal sampling intervals for obtaining information about these parameters. We verify the estimation method through a simulation estimation study, and test each dataset to ensure suitability to the estimation method. We find that the force of infection differs by population prevalence, and the infectious period is estimated to be between 12 and 20 days. We also find that optimal sampling interval depends on setting, with an optimal sampling interval between 9 and 11 days in a high prevalence setting, and 21 and 27 days for a lower prevalence setting. These estimates unlock future model-based investigations on the transmission dynamics of skin sores.

Data is made available in the supporting information. Estimated posterior distributions and the code used to perform simulation/estimation experiments is contained in the supporting information.

Funding: MJL is funded by an Australian Postgraduate Research Training Program scholarship. This work is supported by an NHMRC Project Grant titled 'Optimising intervention strategies to reduce the burden of Group A Streptococcus in Aboriginal Communities' (GNT1098319). We thank the NHMRC Centre for Research Excellence in Infectious Disease Modelling to Inform Public Health Policy (GNT1078068). JMV is supported by an NHMRC Principal Research Fellowship (GNT1117140). SYCT is supported by an NHMRC Career Development Fellowship (GNT1145033). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Impetigo (skin sores) is a condition that remains of public health interest. Late sequelae of acute rheumatic fever and rheumatic heart disease, combined with a high prevalence in remote Australian Aboriginal communities, Fiji, and other areas of socio-economic disad-vantage, mean that impetigo is a substantial contributor to the burden of disease in these settings. Despite decades of study, key quantities of interest from a transmission dynamics perspective—including the force of infection, infectious period and reproductive ratio—have not yet been determined. Such measures are arguably crucial for making informed decisions on future surveillance activities and intervention strategies. Using a series of computational and statistical methods, we find that the infectious period in remote Australian Aboriginal communities is between 12 and 20 days, and that the force of infection varies by setting. Further, we show sampling every 10 days in future surveys is optimal for further refining these estimates.

Introduction

Infections with impetigo (commonly known as skin sores) remain highly prevalent in remote Australian Aboriginal communities, as well as Fiji and areas of socio-economic disadvantage [1, 2]. Skin sore infections in these settings are primarily caused by *Staphylococcus aureus*, and Group A *Streptococcus* (GAS). GAS is associated with post-infectious sequelae such as acute rheumatic fever and rheumatic heart disease, of which Australia has one of the highest recorded prevalences globally [1]. Despite a relatively high level of understanding about the specifics of the GAS bacterium [3–7], comparatively little is known about the natural history of skin sore infection. Furthermore, what is known is often based on historical studies from a prior generation and from a different, non-endemic, geographical region [8–11]. We aim to utilise a dynamic transmission model for skin sores to estimate two key quantities: the *force of infection*, and the *duration of infectiousness*. In the absence of information relating to immunity post-infection, we assume skin sore transmission follows the dynamics of the Susceptible-Infectious-Susceptible (SIS) model. Calculation of these two key quantities will contribute to the development and parameterisation of models which will in turn inform the design of intervention strategies aimed at reducing prevalence.

We analyse three separate datasets, all from remote Australian communities, documenting the infection dynamics of skin sores in individuals. The first dataset consists of public health network presentation data for 404 children under five years of age [12–14], collected as part of the East Arnhem Healthy Skin Project; the second contains longitudinal data for 844 individuals from three rural Australian communities, collected during household visits [15], and the third is comprised of survey visits for 163 individuals who participated in a mass treatment program [16], of which the primary endpoint was control of scabies infection. To analyse these data, we linearise the SIS model about the endemic equilibrium, and derive an expression for the likelihood of the two model parameters. By utilising Markov chain Monte Carlo (MCMC) methods, we obtain estimates of each of the force of infection. Finally, by utilising optimal experimental design, the optimal sampling strategy to inform estimation of these parameters for use in future studies is obtained.

Materials and methods

Ethics statement

Ethics approval for reuse of existing data was obtained from The Human Research Ethics Committee of the Northern Territory Department of Health and Community Services and Menzies School of Health Research (Ethics approval number 2015-2516). Permission was also obtained from the custodians of each dataset. This project has been conducted in association with an Indigenous Reference Group, as well as an ongoing stakeholder group which contains Aboriginal Australian community members.

The Susceptible-Infected-Susceptible model

We consider a stochastic representation of the Susceptible-Infectious-Susceptible (SIS) model [17]. In this model, individuals are either susceptible (*S*) or infectious (*I*). The transition rate from susceptible to infectious, known as the *force of infection*, and denoted $\lambda = \beta I/N$, where β is the transmissibility parameter, is non-linear. This non-linearity is one of the key features of dynamic infectious disease models. However, this means that to model a population of individuals, the state of each individual is required (to know the prevalence, i = I/N). For the SIS model with individuals explicitally stated, the size of the required state space is 2^N [18]. When constructing the Markov chain representation of the SIS model then, the generator matrix, Q, is $2^N \times 2^N$, meaning that for large numbers of individuals, computing the matrix exponential $\exp(Qt)$, is computationally intractable. The result of this, then, is that performing inference with infectious disease models is challenging [19–24].

When the dynamics of the SIS model are at (or close to) equilibrium, then the force of infection, λ , is approximately constant. As such, we approximate the SIS model by a two-state process with a *constant* force of infection. By making this approximation, and assuming individuals are otherwise identical, it follows that a Markov chain consisting of only two states is required, independent of the underlying population size. This approximation has a straightforward likelihood calculation, which allows estimation of parameters in a Bayesian framework and also calculation of the optimal sampling interval for future study designs through the use of optimal experimental design.

Linearisation of the SIS model. The standard SIS model can be described using two transitions, infection and recovery, and two parameters, the transmissibility, β , and the rate of recovery, γ (Table 1). Ignoring demographic processes, the total number of individuals in the population is fixed.

One of the most important quantities in infectious disease modelling is the *basic reproductive ratio*, R_0 , defined as the mean number of secondary infection events caused by a single infectious host, in an otherwise susceptible population. The basic reproductive ratio functions as a threshold parameter, where if $R_0 \le 1$, an outbreak of disease will not occur, while if $R_0 >$ 1, then there is a non-zero probability of a disease outbreak occurring. For the SIS model, the

Table 1. Transitions of the SIS model. The force of infection is given by λ , the transmisibility parameter β and the rate of recovery by γ .

Transition	Rate
$(S, I) \to (S-1, I+1)$	$\lambda \coloneqq \beta I/N$
$(S, I) \to (S+1, I-1)$	γ

https://doi.org/10.1371/journal.pcbi.1007838.t001

basic reproductive ratio is

$$R_0 = \frac{\beta}{\gamma}.$$

The quasi-equilibrium solution of the SIS model is well known [18], and the endemic prevalence of disease is

$$t^* = 1 - \frac{1}{R_0}$$

Let *t* be time of interest of the process. The force of infection, $\lambda(t)$ is defined as,

$$\lambda(t) \coloneqq \beta i(t).$$

At equilibrium, the prevalence is approximately constant, and so the force of infection can be approximated by

$$\lambda(t) = \beta i^* \approx \lambda.$$

By performing this linearisation, it is assumed that the dynamics of disease are and remain at equilibrium. It follows that we may consider a single individual. The generator matrix for the Markov chain for the life-course of that individual is

$$Q = \begin{bmatrix} -\lambda & \lambda \\ & \\ \gamma & -\gamma \end{bmatrix},$$

and the matrix exponential of Q is

$$P(t) = e^{Qt} = \frac{1}{\gamma + \lambda} \begin{bmatrix} \gamma + \lambda e^{-t(\gamma + \lambda)} & \lambda - \lambda e^{-t(\gamma + \lambda)} \\ \gamma - \gamma e^{-t(\gamma + \lambda)} & \lambda + \gamma e^{-t(\gamma + \lambda)} \end{bmatrix}.$$
 (1)

The matrix in Eq (1), combined with an initial state and time *t*, gives the probability distribution for the Markov chain. It is possible to calculate expressions for the equilibrium prevalence, I^* , and the basic reproductive ratio, R_0 , in terms of λ and γ . Solving for the equilibrium distribution of the linearised SIS model

$$\pi Q = 0,$$

gives the equilibrium prevalence

$$i^* = \frac{\lambda}{\lambda + \gamma}.\tag{2}$$

From the standard SIS model, it is also known that the basic reproductive ratio, $R_0 = \beta/\gamma$, and $\lambda = \beta I^*$. It follows that the basic reproductive ratio, R_0 , is given by

$$R_0 = \frac{\beta}{\gamma} = \frac{\lambda}{\gamma i^*},$$

and substituting Eq(2) gives

$$R_0 = \frac{\lambda + \gamma}{\gamma}.$$
(3)

Given these simple closed form expressions for the key quantities of interest, it is possible to perform estimation in a Bayesian setting, using interval-censored data.

Data

Three separate datasets collected in Australian Aboriginal communities are considered in this study: data from public health network presentation (PHN) data on 404 children from birth to five years of age, collected as part of the East Arnhem Healthy Skin Project; data for 844 individuals from three communities, collected during household visits (referred to as the HH dataset); and data from 163 individuals who were observed for over 25 months as part of a mass treatment program in a single rural community (RC). Each dataset consists of longitudinal observations of each individual, where their infection status is recorded at each observation. The times between presentations are heavily right skewed in each dataset, with a median time to next presentation of 9 days for the PHN data, 61 days for the HH data and 119 days for the RC data. The number of observations in total is also highly variable with 13,439 observations in the PHN data, 4,507 in the HH data and 626 in the RC data. Kernel density estimates of the distribution of time until the next presentation, with the observed data overlayed, are shown in Fig 1. The suitability of each of these datasets for inferring the force of infection, λ , rate of recovery, γ , and the basic reproductive ratio, R_0 is investigated in Section Verification of presentation distributions. It is worth noting specifically that the PHN data contains information only on children from birth to five years of age, while the other two datasets contain information on individuals of all ages. Prevalence of skin sores is known to be age-dependent [25] and so by not modelling any age-structure, we are ignoring these differences.

Data structure. Recall that the datasets which are considered consist of longitudinal observations for each individual, with an individual's infection status being noted as either susceptible or infected at each point. The observation is not continuous in nature, with the individual's infection status only being known at each sampling point. Data of this form are known as *interval-censored*, or *panel* data. Interval-censored data are common in epidemiology, and inference in a frequentist setting is well established [26]. Let the state of individual *i* at observation *j* be $X_{i,j}$, and the time at which the *j*th observation is made be $t_{i,j}$. The likelihood for a single individual, *i*, can be evaluated as

$$L_i(\lambda, \gamma) = \prod_j P_{X_{ij}, X_{ij+1}}(t_{i,j+1} - t_{i,j})$$



which is the relevant entry of matrix P in Eq.(1), evaluated at the time difference between

Fig 1. Distribution of (A) time between presentations and (B) age of patients at presentation for each of the three datasets—PHN, HH and RC with the empirical data overlayed.

https://doi.org/10.1371/journal.pcbi.1007838.g001

observations, $t_{i,j+1} - t_{i,j}$. It follows that the likelihood for the entire population is

$$L(\lambda,\gamma) = \prod_{i} L_{i}(\lambda,\gamma).$$
(4)

It is important to note that the likelihood in Eq (4) has assumed what is known as *ignorable* sampling times. That is, the sampling times are chosen independently of the outcome of the process. When sampling times are chosen in advance, as they were in the HH and RC datasets, then the sampling times have been proven to be ignorable [26]. For the PHN data, observations were made under what is termed a *doctor's care* scheme, whereby the next observation time is chosen at the current observation time, and based on an individual's disease state at that time. The sampling times are proven to be ignorable if the following two conditions are true [27]:

- The probability of individual *i* being in a given disease state $u_{i,j}$ at time $t_{i,j}$, given all infection history until this point, $H_{i,j-1}$, is independent of whether an examination is carried out at this time and past examination times, and
- The conditional distribution of the *j*th observation time, $t_{i,j} = P(T_{i,j} = t_{i,j}|H_{i,j-1})$, where $T_{i,j}$ is the random variable representing the time of the *j*th infection for individual *i*, is functionally independent of the transmission parameters.

The first of these conditions effectively means that the infection status at time $t_{i,i}$ only depends on the status at time $t_{i,i-1}$ and on $t_{i,i-1}$ and $t_{i,i}$, but not on whether the individual's infection status is sampled at time $t_{i,i}$ or on previous sampling times. As treatment is prescribed by a doctor's visit, it is possible that this condition is violated. However, the dataset does not contain information on the form of treatment administered meaning that it cannot be assumed that the administered treatment is for skin sores, and almost 60% of presentations to the clinic contain no information on skin sores (and so one could assume that the primary reason for the visit is not skin sores). Further, it is noted that the estimate of infectious period in any modern setting will be reduced by the presence of treatment. As such, it is assumed that the first condition is true. The second condition means that the next observation time is conditionally independent of the transmission process. This condition is assumed to be true here due to the high frequency of presentation in this dataset, even when an individual does not have skin sores. A large number of doctors visits do not contain information about an individual's status with skin sores, including instances where an individual had been marked as infected one day prior. If we ignore these 'missing' entries, the mean time to next positive presentation following a negative presentation is 27.2 days, while the mean time to next negative presentation following a negative presentation is 23.2 days. If skin sores caused more frequent doctors presentations, then we would expect these numbers to be reversed. However, the empirical mean time to next presentation is sensitive to the frequency of missing data, and so it is unclear whether this difference can be attributed to a change in patient behaviour based on infection status, or limitations in data collection. The analysis proceeds on the basis that all sampling schemes in the given data are ignorable, but it is noted that this may not be the case.

It is important to note that in the settings studied here, treatment is routinely offered and applied to skin sores. This will augment the estimate of the recovery rate, γ , to be the average duration of infection in the presence of treatment. However, as treatment is routinely applied in many settings where skin sores are endemic, this estimate is still relevant when considering control schemes and survey designs.

Both the force of infection, λ , and the infectious period, γ are estimated in a Bayesian context using Markov chain Monte Carlo estimation (MCMC). The MCMC is performed using

the No-U-Turn sampler implemented in Stan [28], using 10,000 iterations for 4 chains, for a total of 40, 000 iterations. The code used to perform this estimation is available at http://github.com/MikeLydeamore/TMI/. We use truncated normal distributions, $T\mathcal{N}(0.5, 0.5)$, truncated at 0, as priors for both the force of infection, λ , and the infectious period, γ . To calculate the basic reproductive number, R_0 , we apply the formula in Eq.(3) for each sample from the posterior distribution.

Results

We start by verifying the methodology through the use of a simulation estimation study, whereby individuals are simulated from the linearised SIS model, and we attempt to recover parameters through the estimation routine detailed in the previous section. To explore the suitability of the methods and available datasets, we choose reasonable values for the force of infection, λ , and the rate of recovery, γ . In the main text, we use $\lambda = 1/60$ and $\gamma = 1/20$, while S4 Text presents results for different chosen values. After these verifications, the estimation method is applied to the observational data.

Verification of methodology

There are multiple sources of stochastic variability in this setting, including the underlying population which is observed, the realisation from the observation distribution and the MCMC method itself. The first two of these potential causes for variation are investigated in detail here.

To investigate the variability in the underlying population, the estimation procedure is performed on 64 randomly generated populations from the linearised SIS model, and each of the 400 members of each simulated population are observed once daily for one year. This high frequency of observation means that the only source of meaningful variability is that which comes from the linearised SIS model. The top row of Fig 2 shows the marginal posterior estimates for the force of infection, λ , the rate of recovery, γ , and the basic reproductive ratio, R_0 , for populations simulated using $\lambda = 1/60$ and $\gamma = 1/20$. Each violin plot shows an individual (marginal) posterior distribution for the parameter of interest from a randomly selected population, while the boxplot shows the variability of the posterior mean for each parameter over all 64 realised populations. The within-simulation variability is relatively high, even in this case with daily observation. However, the method estimates each parameter well and in an unbiased manner.

Next, potential variability in the observation distribution is considered. Again, a population of 400 individuals is simulated, and each simulated individual observed at times drawn from the observation distribution obtained from the PHN dataset (shown in Fig_1) over a time horizon of 1 year (Fig_2(B)). It is satisfying that although the sampling interval in the PHN dataset is notably longer than the daily case shown in Fig_2(A), the estimation method is still able to recover the simulated parameters. This suggests that oversampling the population (Fig_2(A)) gives little benefit to estimates of the parameters. Comparatively, it makes sense that if the sampling interval is too large, then no information will be gained. An example of this phenomenon is shown in Fig_3, where 20 samples are made of the population, separated by some sampling interval. The figure shows that a short sampling interval and a relatively short time horizon means that information about the parameters is difficult to recover. Similarly, a long sampling interval increases the variance in the parameter estimates. This suggests that there exists some *optimal* sampling interval. This concept will be returned to in Section *Prospective sampling strategies*.



Fig 2. Marginal posterior distributions for the force of infection, λ , the rate of recovery γ , and the basic reproductive ratio, R_0 , from 8 randomly generated populations from the linearised SIS model under two different observation distributions. The mean of each distribution is given by the white circle. The boxplot at the bottom of each panel represents the means of 64 marginal posteriors. The true value which was used to generate each population is represented by the blue line ($\lambda = 1/60$, $\gamma = 1/20$). The two different observation distributions are (A): Observed daily over 1 year and (B) observed according to the empirical presentation distribution from the PHN data over 1 year. Both observation distributions yield good estimates of the simulated parameters. The observation distribution from the RC dataset was tested, but has not been visualised as the estimates were far from the true values (See Fig 3).

Verification of linearisation procedure

Having established that parameters can be re-estimated from the linearised model, we now look to verify whether the linearisation of the SIS model is valid. To do this, an individual-based implementation of the *full* (non-linearised) SIS model is used. The chosen parameters are $\beta = 0.067$ and $\gamma = 1/20$ (giving $\lambda = 1/60$ and an endemic prevalence of 25%), and 300 individuals. The Markov chain is seeded with 125 infected individuals, which is close to the equilibrium of this system. The system is run for 10 years before observation begins. The population is simulated from the full SIS model, and the estimation is performed using the *linearised* model. No stochastic extinction occurred in any of the simulations throughout this work. Fig.4 shows results from 64 realised populations, under the observation distribution from the PHN dataset. The recovery rate, γ , is estimated accurately and with relatively small variance. The force of infection, λ , is somewhat underestimated on average with a relative error in the mean of 15%, although the variability is large. This underestimate carries over to the estimate of the basic reproductive ratio, R_0 . However, the true parameters are within the 95% confidence interval when averaged over the 64 simulations, similar to that seen in



Fig 3. Variance in the estimates of the force of infection, λ , and the rate of recovery, γ , for a range of sampling intervals. Estimates were performed on 64 realisations of the simulated populations, each with parameters $\lambda = 1/60$ and $\gamma = 1/20$. Each realisation contains 20 observations from the simulated population, leading to the time horizon for each realisation being 20 × sampling interval.

Fig 2 under the same observation distribution. Thus, it is concluded linearisation of the SIS model is valid when the dynamics are near equilibrium.

Verification of presentation distributions

Before estimating the force of infection, λ , and the rate of recovery, γ , for the three datasets discussed, the frequency of presentations must be checked to determine if they are sufficient for use with the method. Fig 5 shows a simulation estimation study using the presentation distributions from the PHN and HH datasets. Both datasets give good estimates. When considering the RC dataset, recall the presentation distributions shown in Fig 1. The RC dataset has a much wider sampling interval compared to the PHN and HH datasets. We suspect that this presentation distribution may not hold sufficient information to recover the parameters of interest. However, as the prevalence is observed at each survey visit, estimating the basic reproductive ratio, R_0 , may still be possible. Fig 6 shows the prior distributions, with samples from



Fig 4. Marginal posterior distributions for the force of infection, λ , the rate of recovery γ , and the basic reproductive ratio, R_0 , from 8 randomly generated populations from the full (non-linearised) SIS model under the empirical observation distribution from the PHN data, over 1 year. The mean of each distribution is given by the white circle. The boxplot at the bottom of each panel represents the means of 64 marginal posteriors. The true value which was used to generate each population is represented by the blue line ($\lambda = 1/60$, $\gamma = 1/20$). The simulated parameters are recovered successfully.

https://doi.org/10.1371/journal.pcbi.1007838.g004



Fig 5. Marginal posteriors for the force of infection, λ , the rate of recovery γ , and the basic reproductive ratio, R_0 , from 8 randomly generated populations from the linearised SIS model under two different observation distributions. The mean of each distribution is given by the white circle. The boxplot at the bottom of each panel represents the means of 64 marginal posteriors. The true value which was used to generate each population is represented by the blue line ($\lambda = 1/60$, $\gamma = 1/20$). The two different observation distributions are taken from the (A): PHN dataset (1 year) and (B) HH dataset (1 year).

the posterior distribution overlayed, under the observation distribution from the PHN dataset (panel A) and the RC dataset (panel B). Under the observation distribution from the PHN data, the posterior distribution samples are tightly clustered, with variance much smaller than in the prior distributions. Indeed, estimates are so localised relative to the prior that the samples appear to be overlayed in the figure. Comparatively, when the observation distribution is that seen in the RC dataset, the posterior samples are strongly correlated with a wide variance, indicating that this dataset does not have sufficient sampling frequency to separately estimate both the force of infection, λ , and the rate of recovery, γ . However, the posterior distribution samples align with the simulated prevalence (and thus the basic reproductive ratio, R_0). The RC dataset can still be used to estimate these quantities.

Having verified the suitability of each of the datasets to this estimation method, the next step is to estimate each of the force of infection, λ , the rate of recovery, γ , the prevalence of disease and the basic reproductive ratio, R_0 .

Estimation from data

For the PHN and HH datasets, relatively similar estimates for the infectious period, $1/\gamma$ (12 days for the PHN dataset, and 20 days for the HH dataset) are obtained. However, notably different estimates for the force of infection, λ , were obtained. In the PHN dataset, the mean



Fig 6. Prior distribution (concentric rings) with 20,000 samples from the posterior distribution (black points) overlayed from a randomly generated population under the observation distribution from (A) the PHN dataset, (B) the HH dataset, and (C) the RC dataset. The red line is the set of parameter values which give the true prevalence in the simulated population. In panels (A) and (B), the samples are tightly clustered with variance far smaller than the prior distribution. In panel (C), the samples are highly correlated, and with high variance, indicating the two parameters of interest cannot be uniquely determined, but their ratio (and so R_0) can.

force of infection is estimated at 1/20.21, while in the HH dataset, the estimate is 1/202.07—an order of magnitude different. This difference follows through to estimates of the basic reproductive ratio, R_0 (1.60 vs 1.10), and the prevalence, estimated to be 37.5% in the PHN dataset and only 9% in the HH dataset. For the RC dataset, R_0 is estimated to be 1.42, and the prevalence to be 26.9%. Point estimates of prevalence in all three study locations have been reported previously (Table 2) [15, 16, 29], at 35.6% in the region in which the PHN dataset was collected, 13.1% in the region where the HH dataset was collected and 35% in the region where the RC dataset was collected. These prevalence estimates align well with the estimates obtained using our method.

Prospective sampling strategies

Thus far, the focus has been on previously collected datasets from which to estimate parameters. If the sole aim of a study was to collect data to *best* estimate these parameters, then the natural question to ask is when should individuals be sampled? Aided by the simple structure of the linearised SIS model, this question may be answered through optimal experimental design [30]. We take the approach of robust optimal experimental design, under the ED-optimality criterion [31, 32]. Let $\boldsymbol{\delta} = (\delta_1, \dots, \delta_{n-1})$ define an n-sampling design with spacing δ_i , $i = 1, \dots,$

Dataset	Parameter [units]	Mean	95% CI
PHN	Force of infection (λ) [1/days] 0.049		(0.042, 0.059)
	Infectious period (γ^{-1}) [days]	12.19	(10.23, 14.55)
	R ₀	1.60	(1.56, 1.65)
	Prevalence	37.5%	(31.0, 39.4)
Literature [29]	Prevalence 35.6% (32		(32.9, 38.3)
HH	Force of infection (λ) [1/days]	0.0049	(0.0040, 0.0062)
	Infectious period (γ^{-1}) [days] 19.97		(16.19, 24.56)
	R ₀	1.10	(1.09, 1.11)
	Prevalence	9.1%	(8.3, 10.0)
Literature [15]	Prevalence 13.1% No		Not provided
RC	R ₀		(1.34, 1.51)
	Force of infection (λ)	_	Not identifiable
	Infectious period (γ^{-1})	_	Not identifiable
	Prevalence	29.6%	(25.4, 33.8)
Literature [16]	Prevalence	35%	Not provided

Table 2. Parameter estimates for the force of infection, λ_2 , and the infectious period, $1/\gamma$ from the three different datasets. Note this method estimates the rate of recovery, γ , but the infectious period is reported here for clarity.

n-1 between subsequent observations. Then, the optimal sampling design, δ^* , is given by

$$\boldsymbol{\delta}^* = \operatorname*{argmax}_{\delta} \int \det \left(I(\boldsymbol{\delta}, \boldsymbol{\theta}) \right) p(\boldsymbol{\theta}) d\boldsymbol{\theta}, \tag{5}$$

where $\theta = \{\lambda, \gamma\}$, $I(\delta, \theta)$ is the Fisher Information matrix, det is the determinant operator, and $p(\theta)$ is the prior distribution. Note that the optimal sampling interval, δ^* , is dependent on the prior distribution, $p(\theta)$. Two designs are considered for each dataset. The first is termed the *variable* sampling interval, where the *i*th sampling interval, δ_i , is unrestricted, and n = 11 design spacings are chosen. Although this design strategy is optimal over a 12 visit design, adhering to the varying intervals may be difficult from an implementation perspective. A more practical strategy, and the second considered here, is termed the *fixed* sampling interval, where $\delta_i = \delta$, $\forall i$. This is equivalent to considering n = 1 design spacing, as the population dynamics are assumed to be at equilibrium throughout the study.

The integral in Eq (5) is approximated using a Monte Carlo estimate with 5,000 samples. Each individual is observed 12 times. We use the induced natural selection heuristic for finding optimal strategies [33]. For detail on the algorithm inputs and evidence of convergence, see S3 Text.

Recommended sampling strategies. We calculate the optimal strategy using the posterior distributions obtained from the PHN dataset, HH dataset and the union of the these two posterior distributions as the prior distribution in Eq (5). The results for both the variable interval strategy and the fixed interval strategy are shown in Table 3.

Under the constraint of equal observation intervals, and restricted to whole days any sampling interval between 9 days and 11 days gives a Fisher Information within 97% of the maximum for the PHN dataset. Comparatively, for the HH dataset, any sampling interval between 21 days and 28 days gives a Fisher Information within 97% of the maximum. Combining the two posterior distributions, any sampling interval between 21 and 28 days is within 97% of the maximum. However, it should be noted that a sampling interval of 23.4 days achieves only 30% of the maximum Fisher information possible in the PHN dataset, but 99% of the

Data Source	Interval	Optimal Design (days)
PHN	Variable	(12.9, 12.3, 10.4, 12.0, 10.3, 12.5, 12.5, 9.1, 11.3, 10.6, 11.6, 13.2)
	Fixed	9.9
HH	Variable	(18.1, 23.0, 30.5, 31.1, 31.6, 27.8, 30.0, 31.7, 30.0, 30.6, 32.6, 29.4)
	Fixed	24.2
Combined	Variable	(16.2, 23.5, 24.3, 32.0, 28.2, 26.6, 31.6, 29.2, 28.2, 28.1, 30.8, 33.9)
	Fixed	23.4

Table 3. Optimal sampling strategies (in days) using the posterior distributions obtained from the PHN and HH datasets, as well as the union of these two posterior distributions. Two sampling strategies are considered: variable, where the time between each observation is allowed to vary, and fixed.

https://doi.org/10.1371/journal.pcbi.1007838.t003

maximum in the HH dataset. This highlights the importance of specifying the optimal sampling strategy according to the specific scenario.

Interestingly, the optimal design spacing for the fixed strategy is *not* the minimum of the optimal design spacing for the variable strategy. We propose the following hypothesis for this phenomenon: when the observation interval is allowed to vary, we can effectively 'spend' a single observation close to the previous in order to potentially gain a lot of information. However, in the fixed interval strategy, this option is not available, and so to avoid 'wasting' observations, a more conservative strategy becomes the optimal.

To understand the difference in the optimal sampling times, recall that the expression in Eq (5) maximises the Fisher Information, which through the Cramer–Rao lower bound, can be thought of as minimising the variance of the parameter estimates [34]. This estimate inherently depends on the underlying parameters of the system: when events (i.e., infection and recovery) are happening slowly (i.e., low prevalence) then sampling should happen less often, while when events are happening frequently (i.e., high prevalence), then sampling should happen more often. In the case where little prior information about the system is available, then it may be more appropriate to adopt a 'conservative' sampling strategy, which here is the faster of the two presented strategies. Doing this yields a Fisher Information of 53% of the maximum for the HH dataset. The conservative strategy is presented in <u>S4 Text</u>. Overall, the conservative strategy generally gives good estimation accuracy (up to 10% error in a simulation-estimation study), and so is a viable 'catch-all' strategy in the absence of prior information such as the prevalence.

Discussion

We have provided the first model-based estimates for the duration of a skin sore infection (between 12 and 20 days), the force of infection and basic reproductive ratio (1.1 to 1.6) in three different settings. Furthermore, the optimal sampling interval for future strategies has been determined, assuming that a study's primary goal is to estimate the force of infection and duration of infectiousness. By performing the estimation in a modelling framework, the interval-censored nature of the data has been incorporated. Although the frequentist version of this estimation technique has been utilised in other disease settings [35, 36], to our knowledge this is the first time these quantities have been computed for skin sores.

Previous work on the duration of skin sore infection has estimated that under treatment, skin sore infections clear in approximately 50% of individuals in 2 days, and 85% of individuals in 7 days [37]. In this study, we do not have information on whether an individual was prescribed treatment, but it is expected that a proportion of the population were prescribed and used antibiotics. For the remainder of the population who were not treated with antibiotics, it is expected that their clearance time would be longer. As our data are a combination of treated

and untreated individuals, we propose that an exponential distribution for the duration of infectiousness is reasonable. Further, without more frequent observations, we are unlikely to be able to distinguish between other proposed distributions with confidence (<u>S5 Text</u>). Should accurately characterising the distribution of the duration of infectiousness be of interest, a similar approach to that in Section *Prospective sampling strategies* can be used to design a study to best discriminate between different models [<u>38</u>].

These results have been calculated using a linearised SIS model, in which the transmission rate has been assumed to be constant, and disease dynamics are at equilibrium. This assumption has allowed some simple analytic results which are often not able to be determined for traditional infectious disease dynamic models. However, it is important to note that the assumption of equilibrium dynamics is likely to be violated in real-world settings, particularly in the event of mass drug administration. Mass drug administration has been implemented in these communities in the past [29, 39], and was ongoing during the period of data collection in the RC dataset, although skin sores was not the primary outcome of the program in the RC setting [16]. It is also important to note that the SIS model structure, by construction, does not incorporate any period of immunity, or other potential disease states. Carriage (i.e. infected but not showing symptoms), in particular has been demonstrated for skin sore infections in the past [11, 15] and inclusion of carriage in models has been shown to substantially change predicted intervention outcomes [40]. Given the hyperendemic prevalence of skin sores in this setting, the observed high infectiousness of skin to skin transmission, and in the absence of longitudinal data related to carriage available in this study, we have ignored the carrier state in this model. Extension in this area represents an important path for future work.

It must also be noted that the condition of skin sores can be caused by a number of pathogens. The microbiology of skin sores in the Northern Territory, Australia, has been studied previously [41]. *Streptococcus pyogenes* remained the dominant pathogen, but co-infection with *Staphylococcus aureus* was present. Without microbiologic information present in our data sets, we are unable to determine which pathogen is causing infection. Accordingly, we see these results as a quantification of skin sores as a general condition.

In the populations in which these data were collected, treatment is routinely administered for skin sores. Thus, these estimates of the infectious period are inclusive of the effect of treatment, and so are likely to be lower than the *natural* infectious period (that is, in the absence of treatment being available). Although this interpretation of the infectious period is different to the *natural* infectious period, it is arguably more useful in an epidemiological context, as treatment will be given in any modern setting for a skin sores infection.

There are a number of key differences between the three datasets considered. The PHN dataset only has observations of children under five years of age. Extrapolation from this dataset to the entire population should be performed with caution as the prevalence of skin sores appears to be age-specific [25], and the average age of participants in the PHN data was younger than in the other two datasets. Despite this demographic difference, the relative similarity of the estimates of the infectious period from the HH data (in which the general population was studied) does provide some reassurance of the estimated numbers. Further, sampling times in the PHN dataset were not fixed in advance, but were rather driven by patients or health professionals. It has been assumed these sampling times are ignorable, but further investigation into this assumption may be warranted.

As well as estimation of key parameters for models of skin sores transmission, information about future experimental designs has also been provided. Although the optimal sampling interval is a function of both the force of infection and the infectious period, being able to calculate this interval provides helpful information to improve the efficiency of future study designs, or evaluation of disease control programs. These parameter estimates unlock future model-based investigations for skin sores. By providing estimates for both the force of infection and the duration of infectiousness, more complex models which include covariates such as scabies, non-homogenous contact patterns, and population mobility can be considered, and the impact of treatment strategies in these settings can be evaluated. It is our hope that these models will lead to the development of innovative disease control measures, the application of which will reduce the burden of skin disease and health inequalities.

Supporting information

S1 Text. MCMC diagnostics. MCMC diagnostics relating to convergence of the posterior distributions.

(PDF)

S2 Text. Derivation of the Fisher Information matrix. Derivation of the Fisher Information matrix for both variable and constant time between observations. (PDF)

S3 Text. Optimal sampling strategy diagnostics. Diagnostics of the optimisation of the sampling strategies.

(PDF)

S4 Text. Conservative sampling strategy. Utility of the 'conservative' sampling strategy compared to the optimal sampling strategy. (PDF)

S5 Text. Model sensitivity. Simulation study where the simulated data comes from a model with two infectious phases but is estimated using a single infectious phase. (PDF)

S1 File. Patient ages at presentation. Patient ages at presentation, separated by dataset in which they appeared.

(CSV)

S2 File. Posterior distributions for the three datasets. Posterior distributions for the force of infection and infectious period from the three datasets. (ZIP)

S3 File. R code for performing simulation/estimation and verifying Fisher Information. R code for performing simulation/estimation experiments, and numerically verifying the Fisher Information expressions included in the TMI package. (ZIP)

Acknowledgments

We would like to thank the Menzies School for Health Research and related project staff, for providing the data from the East Arnhem Healthy Skin Project, staff at the primary health care centres and the members of the remote indigenous communities for their participation. We acknowledge our partners in this work: Northern Territory Remote Health, Aboriginal Medical Services Alliance Northern Territory, Northern Territory Centre for Disease Control, One Disease and Miwatj Health and the NHMRC funded HOT NORTH initiative. We acknowledge the Lowitja Institute and the Cooperative Research Centre for Aboriginal Health who originally funded and lent significant support to the East Arnhem Healthy Skin Project.

Author Contributions

- **Conceptualization:** Michael J. Lydeamore, Patricia T. Campbell, Jodie McVernon, Steven Y. C. Tong, James M. McCaw.
- **Data curation:** Yue Wu, Adrian J. Marcato, Will Cuningham, Jonathan R. Carapetis, Ross M. Andrews, Malcolm I. McDonald.

Formal analysis: Michael J. Lydeamore, David J. Price.

Investigation: Michael J. Lydeamore.

Methodology: Michael J. Lydeamore, David J. Price.

Project administration: Steven Y. C. Tong.

Software: Michael J. Lydeamore.

Supervision: Patricia T. Campbell, Jodie McVernon, James M. McCaw.

Writing - original draft: Michael J. Lydeamore.

Writing – review & editing: Michael J. Lydeamore, Patricia T. Campbell, David J. Price, Yue Wu, Adrian J. Marcato, Will Cuningham, Jonathan R. Carapetis, Ross M. Andrews, Malcolm I. McDonald, Jodie McVernon, Steven Y. C. Tong, James M. McCaw.

References

- Bowen AC, Mahé A, Hay RJ, Andrews RM, Steer AC, Tong SYC, et al. The Global Epidemiology of Impetigo: A Systematic Review of the Population Prevalence of Impetigo and Pyoderma. PLOS ONE. 2015; 10(8):e0136789. https://doi.org/10.1371/journal.pone.0136789 PMID: 26317533
- Romani L, Steer AC, Whitfeld MJ, Kaldor JM. Prevalence of Scabies and Impetigo Worldwide: A Systematic Review. The Lancet Infectious Diseases. 2015; 15(8):960–967. <u>https://doi.org/10.1016/S1473-3099(15)00132-2</u>
- Walker MJ, McArthur JD, McKay F, Ranson M. Is Plasminogen Deployed as a Streptococcus Pyogenes Virulence Factor? Trends in Microbiology. 2005; 13(7):308–313. <u>https://doi.org/10.1016/j.tim.2005.05.</u> 006
- Ong CIY, Gillen CM, Barnett TC, Walker MJ, McEwan AG. An Antimicrobial Role for Zinc in Innate Immune Defense Against Group A Streptococcus. The Journal of Infectious Diseases. 2014; 209 (10):1500–1508. https://doi.org/10.1093/infdis/jiu053
- Rodríguez-Ortega MJ, Norais N, Bensi G, Liberatori S, Capo S, Mora M, et al. Characterization and Identification of Vaccine Candidate Proteins through Analysis of the Group A *Streptococcus* Surface Proteome. Nature Biotechnology. 2006; 24(2):191–197. <u>https://doi.org/10.1038/nbt1179</u> PMID: 16415855
- Cole JN, Barnett TC, Nizet V, Walker MJ. Molecular Insight into Invasive Group A Streptococcal Disease. Nature Reviews Microbiology. 2011; 9(10):724–736. https://doi.org/10.1038/nrmicro2648
- Hysmith ND, Kaplan EL, Cleary PP, Johnson DR, Penfound TA, Dale JB. Prospective Longitudinal Analysis of Immune Responses in Pediatric Subjects After Pharyngeal Acquisition of Group A Streptococci. Journal of the Pediatric Infectious Diseases Society. 2017; 6(2):187–196. https://doi.org/10. 1093/jpids/piw070
- Ferrieri P, Dajani AS, Wannamaker LW, Chapman SS. Natural History of Impetigo. I. Site Sequence of Acquisition and Familial Patterns of Spread of Cutaneous Streptococci. The Journal of Clinical Investigation. 1972; 51(11):2851–2862.
- Dajani AS, Ferrieri P, Wannamaker LW. Natural History of Impetigo: II. Etiologic Agents and Bacterial Interactions. Journal of Clinical Investigation. 1972; 51(11):2863–2871. https://doi.org/10.1172/ JCI107109
- Maddox JS, Ware JC, Dillon HC. The Natural History of Streptococcal Skin Infection: Prevention with Topical Antibiotics. Journal of the American Academy of Dermatology. 1985; 13(2):207–212. https:// doi.org/10.1016/S0190-9622(85)70160-0

- Dudding BA, Burnett JW, Chapman SS, Wannamaker LW. The Role of Normal Skin in the Spread of Streptococcal Pyoderma. The Journal of Hygiene. 1970; 68(1):19–28. <u>https://doi.org/10.1017/</u> S0022172400028461
- Kearns T, Clucas D, Connors C, Currie BJ, Carapetis JR, Andrews RM. Clinic Attendances during the First 12 Months of Life for Aboriginal Children in Five Remote Communities of Northern Australia. PLOS ONE. 2013; 8(3):1–5.
- Clucas DB, Carville KS, Connors C, Currie BJ, Carapetis JR, Andrews R. Disease Burden and Health-Care Clinic Attendances for Young Children in Remote Aboriginal Communities of Northern Australia. Bulletin of the World Health Organization. 2008; 86(4):275–281. https://doi.org/10.2471/BLT.07.043034
- McMeniman E, Holden L, Kearns T, Clucas DB, Carapetis JR, Currie BJ, et al. Skin Disease in the First Two Years of Life in Aboriginal Children in East Arnhem Land. Australasian Journal of Dermatology. 2011; 52(4):270–273. https://doi.org/10.1111/j.1440-0960.2011.00806.x PMID: 22070701
- McDonald MI, Towers RJ, Andrews RM, Benger N, Currie BJ, Carapetis JR. Low Rates of Streptococcal Pharyngitis and High Rates of Pyoderma in Australian Aboriginal Communities Where Acute Rheumatic Fever Is Hyperendemic. Clinical Infectious Diseases. 2006; 43(6):683–689. https://doi.org/10. 1086/506938
- 16. Carapetis JR, Connors C, Yarmirr D, Krause V, Currie BJ. Success of a Scabies Control Program in an Australian Aboriginal Community. The Pediatric Infectious Disease Journal. 1997; 6:494–499.
- Kermack WO, McKendrick AG. A Contribution to the Mathematical Theory of Epidemics. Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences. 1927; 115:700– 721.
- 18. Keeling MJ, Rohani P. Modeling Infectious Diseases in Humans and Animals. Princeton University Press; 2008.
- O'Neill P, Roberts G. Bayesian Inference for Partially Observed Stochastic Epidemics. Journal of the Royal Statistical Society: Series A (Statistics in Society). 1999; 162(1):121–129. https://doi.org/10. 1111/1467-985X.00125
- Black AJ, Geard N, McCaw JM, McVernon J, Ross JV. Characterising Pandemic Severity and Transmissibility from Data Collected during First Few Hundred Studies. Epidemics. 2017; 19(Supplement C):61–73.
- Moss R, Zarebski A, Dawson P, McCaw JM. Forecasting Influenza Outbreak Dynamics in Melbourne from Internet Search Query Surveillance Data. Influenza and other respiratory viruses. 2016; 10 (4):314–323. https://doi.org/10.1111/irv.12376
- 22. Longini IM, Koopman JS, Haber M, Cotsonis GA. Statistical Inference for Infectious Diseases: Risk-Specific Household and Community Transmission Parameters. American Journal of Epidemiology. 1988; 128(4):845–859. https://doi.org/10.1093/oxfordjournals.aje.a115038
- Zarebski AE, Dawson P, McCaw JM, Moss R. Model Selection for Seasonal Influenza Forecasting. Infectious Disease Modelling. 2017; 2(1):56–70. https://doi.org/10.1016/j.idm.2016.12.004
- Nsoesie EO, Brownstein JS, Ramakrishnan N, Marathe MV. A Systematic Review of Studies on Forecasting the Dynamics of Influenza Outbreaks. Influenza and Other Respiratory Viruses. 2014; 8 (3):309–316. https://doi.org/10.1111/irv.12226
- 25. Romani L, Koroivueta J, Steer AC, Kama M, Kaldor JM, Wand H, et al. Scabies and Impetigo Prevalence and Risk Factors in Fiji: A National Survey. PLOS Neglected Tropical Diseases. 2015; 9: e0003452. https://doi.org/10.1371/journal.pntd.0003452 PMID: 25738499
- Jackson C. Multi-State Models for Panel Data: The Msm Package for R. Journal of Statistical Software. 2011; 38(8):28.
- Gruger J, Kay R, Schumacher M. The Validity of Inferences Based on Incomplete Observations in Disease State Models. Biometrics. 1991; 47(2):595–605. https://doi.org/10.2307/2532149
- 28. Stan Development Team. RStan: The R Interface to Stan. 2018;.
- Andrews RM, Kearns T, Connors C, Parker C, Carville K, Currie BJ, et al. A Regional Initiative to Reduce Skin Infections amongst Aboriginal Children Living in Remote Communities of the Northern Territory, Australia. PLOS Neglected Tropical Diseases. 2009; 3(11):1–9. https://doi.org/10.1371/ journal.pntd.0000554
- 30. Fisher RA. The Design of Experiments. 1935;.
- **31.** Pagendam DE, Pollett PK. Locally Optimal Designs for the Simple Death Process. Journal of Statistical Planning and Inference. 2010; 140(11):3096–3105. https://doi.org/10.1016/j.jspi.2010.04.017
- Pronzato L, Walter E. Robust Experiment Design via Stochastic Approximation. Mathematical Biosciences. 1985; 75(1):103–120. https://doi.org/10.1016/0025-5564(85)90068-9

- Price DJ, Bean NG, Ross JV, Tuke J. An Induced Natural Selection Heuristic for Finding Optimal Bayesian Experimental Designs. Computational Statistics & Data Analysis. 2018; 126:112–124. <u>https://doi.org/10.1016/j.csda.2018.04.011</u>
- 34. Casella G, Berger RL. Statistical Inference. vol. 2. Duxbury Pacific Grove, CA; 2002.
- 35. Grassly NC, Ward ME, Ferris S, Mabey DC, Bailey RL. The Natural History of Trachoma Infection and Disease in a Gambian Cohort with Frequent Follow-Up. PLOS Neglected Tropical Diseases. 2008; 2 (12):e341. https://doi.org/10.1371/journal.pntd.0000341
- Ivanek R, Österberg J, Gautam R, Lewerin SS. Salmonella Fecal Shedding and Immune Responses Are Dose- and Serotype- Dependent in Pigs. PLOS ONE. 2012; 7(4):e34660. https://doi.org/10.1371/ journal.pone.0034660
- Bowen AC, Tong SYC, Andrews RM, O'Meara IM, McDonald MI, Chatfield MD, et al. Short-Course Oral Co-Trimoxazole versus Intramuscular Benzathine Benzylpenicillin for Impetigo in a Highly Endemic Region: An Open-Label, Randomised, Controlled, Non-Inferiority Trial. The Lancet. 2014; 384(9960):2132–2140. https://doi.org/10.1016/S0140-6736(14)60841-2
- Hainy M, Price DJ, Restif O, Drovandi C. Optimal Bayesian design for model discrimination via classification. arXiv preprint arXiv:180905301. 2018;.
- 39. Kearns TM, Speare R, Cheng AC, McCarthy J, Carapetis JR, Holt DC, et al. Impact of an Ivermectin Mass Drug Administration on Scabies Prevalence in a Remote Australian Aboriginal Community. PLOS Neglected Tropical Diseases. 2015; 9(10):e0004151. <u>https://doi.org/10.1371/journal.pntd.0004151</u> PMID: 26516764
- 40. Chisholm RH, Campbell PT, Wu Y, Tong SYC, McVernon J, Geard N. Implications of Asymptomatic Carriers for Infectious Disease Transmission and Control. Open Science. 2018; 5(2):172341.
- Bowen AC, Tong SY, Chatfield MD, Carapetis JR. The Microbiology of Impetigo in Indigenous Children: Associations between Streptococcus Pyogenes, Staphylococcus Aureus, Scabies, and Nasal Carriage. BMC Infectious Diseases. 2014; 14(1):727. https://doi.org/10.1186/s12879-014-0727-5

University Library



A gateway to Melbourne's research publications

Minerva Access is the Institutional Repository of The University of Melbourne

Author/s:

Lydeamore, MJ; Campbell, PT; Price, DJ; Wu, Y; Marcato, AJ; Cuningham, W; Carapetis, JR; Andrews, RM; McDonald, M; McVernon, J; Tong, SYC; McCaw, JM

Title:

Estimation of the force of infection and infectious period of skin sores in remote Australian communities using interval-censored data

Date:

2020-10-01

Citation:

Lydeamore, M. J., Campbell, P. T., Price, D. J., Wu, Y., Marcato, A. J., Cuningham, W., Carapetis, J. R., Andrews, R. M., McDonald, M., McVernon, J., Tong, S. Y. C. & McCaw, J. M. (2020). Estimation of the force of infection and infectious period of skin sores in remote Australian communities using interval-censored data. PLoS Computational Biology, 16 (10), https://doi.org/10.1371/journal.pcbi.1007838.

Persistent Link: http://hdl.handle.net/11343/251658

File Description: published version License: CC BY